

Citation:

Zelikson, D and Shubita, M and Wu, J (2020) Adoption of Global Investment Performance Standards: Case of ASEAN. Cogent Business and Management, 7 (1). ISSN 2331-1975 DOI: https://doi.org/10.1080/23311975.2020.1746169

Link to Leeds Beckett Repository record: https://eprints.leedsbeckett.ac.uk/id/eprint/6659/

Document Version: Article (Published Version)

Creative Commons: Attribution 4.0

Deposited by Blomfield, Helen on behalf of Shubita, Moade.

The aim of the Leeds Beckett Repository is to provide open access to our research, as required by funder policies and permitted by publishers and copyright law.

The Leeds Beckett repository holds a wide range of publications, each of which has been checked for copyright and the relevant embargo period has been applied by the Research Services team.

We operate on a standard take-down policy. If you are the author or publisher of an output and you would like it removed from the repository, please contact us and we will investigate on a case-by-case basis.

Each thesis in the repository has been cleared where necessary by the author for third party copyright. If you would like a thesis to be removed from the repository or believe there is an issue with copyright, please contact us on openaccess@leedsbeckett.ac.uk and we will investigate on a case-by-case basis.





#### ACCOUNTING, CORPORATE GOVERNANCE & BUSINESS ETHICS RESEARCH ARTICLE

# Adoption of global investment performance standards: Case of ASEAN

Deborah Zelikson, Moade Fawzi Shubita and Junjie Wu

Cogent Business & Management (2020), 7: 1746169





Received: 01 November 2019 Accepted: 13 March 2020

\*Corresponding author: Moade Fawzi Shubita, Accounting and Finance Subject Group, Leeds Business School, Leeds Beckett University, Rose Bwol 417, Leeds, UK E-mail: m.f.shubita@leedsbeckett.ac.uk

Reviewing editor: Collins G. Ntim, Accounting, University of Southampton, Southampton, UK

Additional information is available at the end of the article

### ACCOUNTING, CORPORATE GOVERNANCE & BUSINESS ETHICS | RESEARCH ARTICLE

## Adoption of global investment performance standards: Case of ASEAN

Deborah Zelikson<sup>1</sup>, Moade Fawzi Shubita<sup>2\*</sup> and Junjie Wu<sup>,2</sup>

**Abstract:** Research on voluntary compliance with accepted international standards has paid overwhelming attention to financial reporting standards, but not to investment performance standards. Previous research on the adoption of the Global Investment Performance Standards has overlooked the unique region of the Association of Southeast Asian Nations. Using 17 years (1999 to 2015) worth of data from all ten countries, which generates 170 country-year observations for each variable of the study, this paper evaluates whether, and how, social and economic pressures influence the adoption of GIPS in the region in the Institutional Theory lens. The results suggest that social pressure is more impactful than economic pressure on the adoption of GIPS. The findings have generated useful contributions and implications in this vein, and several future research directions have been identified.

Subjects: Economics; Finance; Business, Management and Accounting



Deborah Zelikson

#### ABOUT THE AUTHORS

The research team has combined professional and academic dimensions. Deborah Zelikson is the first author of the paper, a professional who has decided to conduct her PG studies, including a PG dissertation, motivated by her professional background. The dissertation was jointly supervised by Junjie Wu and Moade Shubita. This research has benefited from Deborah's outstanding dissertation and has been developed jointly by combining the expertise of the three authors.

Moade Fawzi Shubita who is the corresponding author for this paper has published more than 30 articles, mainly in Finance and Accounting journals, in Accounting and Finance topics, and his research interests cover various directions around market-based accounting research areas, primarily using quantitative research methods.

Junjie Wu is a reader of Finance and has published 50 scholarly articles with nationally and internationally excellent journals. Her multidisciplinary research areas span across accounting, finance, economics, energy, international business and education, using diversified research methods.

#### PUBLIC INTEREST STATEMENT

In a dynamic business environment, uncertainty is the dominant factor, and the need for transparency is no longer an optional dimension as it becomes a necessity by default. Our study contributes to the professional business environment in addressing Global Investment Performance Standards by the Chartered Financial Analyst, as globally accepted standards as proxy for best practice. We have addressed the ASEAN region using country level data, in the hope that our study is used by professionals in ASEAN, as well as elsewhere, in order to motivate the adoption of global standards to help in enhancing sustainability around the globe by showing that there is a social need for this adoption on a voluntary basis with globally accepted investment standards.





Keywords: Global Investment Performance Standards (GIPS); ASEAN; investment profession; regional integration; voluntary adoption; sustainability reporting; Chartered Financial Analyst (CFA) Jel classifications: G11; G15; N95; P17

#### 1. Introduction

Investment management professionals manage various assets, such as shares, bonds and real estate, in order to meet specified investment goals for the benefit of the investors. Nowadays, the investment process has become increasingly complex and international, following increases in investment types and the volume of financial entities, and has demanded the standardisation of investment performance calculation methodology, and a presentation of the returns generated. The adoption of a global standard allows investors to guickly compare the investment performance among managers on a "like for like" basis. As a result, firms that are based in countries with limited or no investment performance standards can comply with a globally accepted format in order to enable them to compete for clients on an equal footing with firms from countries with established practices. Besides, firms from countries with developed standards would have more confidence in a fair comparison against local firms in countries with no previous record of standards (CFA Institute, 2014). As such, the adoption of a voluntary standard for investment performance measurement is described as a "global passport", because firms can transport their reports around the world without having to recalculate figures based on local rules (Evans, 2013). The standards also facilitate a dialogue between the client and the firm with regards to the investment performance reports.

The Global Investment Performance Standards (GIPS) are a set of voluntary and ethical principles for investment managers. They have industry-wide application and contain guidelines on how to calculate and present investment performance to clients, adopted by 85% of the top-100 global asset management firms as of December, 2015 (Grover, 2017), representing 61% of the world's assets under management. At country level, 40 countries have adopted GIPS, as of 1 April 2017 (GIPS, n.d.).

Studies on voluntary reporting standards (for example, International Financial Reporting Standards—IFRS) suggest that there is a correlation between adoption and the influence of social pressures (Darus et al., 2013; Lasmin, 2011; Martinez-Ferrero & Garcia-Sanchez, 2017). However, similar studies are rare on the adoption of investment adoption performance standards (for example, GIPS).

Arguably, the concept of a global standard implies one standard fits all countries. On that basis, it assumes that the environment (for example, the composition of organizations, strategies, customs and laws) and the influences upon the environment, for all countries, are the same. Another assumption is that the perceived economic benefits from the adoption of GIPS, from one country to another, would also be the same. However, due to different socioeconomic and political factors, this relationship does not hold for all countries and regions (Lasmin, 2011). For example, the World Development Indicators show that 37% of the countries that have currently adopted GIPS are classified as developing, and 63% as developed countries (The World Bank, 2016). A counter-example is that the European Union (EU) and the ASEAN are two significantly different regional economic organizations (Martell, 2010); however, they have the same 40% rate of GIPS country adoption (GIPS, n.d.). This observation suggests that the economic drivers may not solely determine the adoption of GIPS for a given country, and alternate factors could also influence GIPS adoption, such as the level of sophistication of education programmes, technical challenges, local regulatory bodies for endorsement of standards, and such like. Also, in terms of the profile of an investment management firm or country of origin, its environment can be directly influenced by sociological factors, including religion, ethnicity, economic status, education, location, and political systems. As pointed out by Friedland and Alford (1991, p. 232), "It is not possible to understand individual or organizational behaviour without locating it in a social context".

Since its foundation in 1967, ASEAN has been in the process of regional integration, but the integration has not yet led to tangible results because of the fact that the region fails to address the necessity for legal change (Orcalli, 2017) due to significant diversities across member countries in local preferences, cultural sensitivities, language, and religion (Vinayak et al., 2014). From an investment management perspective, the diversities suggest that a one-size-fits-all strategy might not be appropriate across the varied markets. However, an alternative argument is that if some ASEAN members have adopted an international standard, such as GIPS, then the adopting countries in the region will influence and encourage the non-adopters in order to consider GIPS compliance. This argument links to the Institutional Theory, which suggests that over time, firms within a field become isomorphic, that is, similar in terms of structure, processes, and the way they react to social influences (DiMaggio & Powell, 1983; Nyahas et al., 2017; Tolbert & Zucker, 1983). Similar studies have leveraged DiMaggio and Powell (1983) three categories of isomorphic change mechanism (coercive, mimetic and normative forces) in order to represent social pressures. Coercive pressures relate to formal and informal influence by legal and regulatory bodies; mimetic effects occur in times of uncertainty, and normative pressure stems from the professionalization of a field.

The influence of social pressures on the adoption of investment performance standards is a topic that remains overlooked.

This study, therefore, examines the impacts of social and economic pressures on the adoption of GIPs in the ASEAN region. The analysis is to achieve the following: (1) determining whether social pressures, represented by coercive, mimetic and normative isomorphic change predictors impact the adoption of GIPS; (2) examining whether economic pressures impact the adoption of GIPS; (3) ascertaining whether social pressures are greater than economic pressures on the influence of a country's adoption of the GIPS; and (4) establishing within social pressures which component is more influential than others in the adoption of GIPS. By achieving these objectives, this study would be the first research of its kind to be conducted in the region on this topic. We are determined to shed light on the issues related, such as the current status of the adoption of GIPS, the roles that social and economic pressures have played in the adoption of GIPS and why, and what the region's policymakers should do to improve the adoption rate, and what future research can be carried out in order to deepen the understanding of this topic.

The rest of the paper is structured as follows: Section 2 introduces the adoption of GIPS in the context of the ASEAN. Section 3 provides a theoretical framework for sustainability reporting. Section 4 establishes the linkage between the empirical literature review and the hypotheses. Section 5 introduces the data and analysis model. Section 6 presents the results, and the final section (Section 7) summarises and concludes the paper.

#### 2. Global investment performance standards within the ASEAN context

The Association of Southeast Asian Nations (ASEAN) was established in 1967 by the original five members (ASEAN-5), Indonesia, Philippines, Singapore, Thailand and Malaysia, and has now extended to ten member countries with a population of more than 650 million. It aims to integrate the region as a whole with the motto "One Vision, One Identity, One Community" in order to achieve the sustainable development of this region, and to compete in the world. After decades of effort, ASEAN has become an investment icon within global financial markets, and is the fourth largest trading region (Thampanya et al., 2020).

Regarding GIPS adoption, the ASEAN has a mix of status: there are four fully adopted GIPS countries of the ASEAN-5, excluding Malaysia. Registered interest countries include Malaysia and Brunei Darussalam, as well as the non-adopters (Cambodia, Lao PDR, Myanmar, and Vietnam) (GIPS, n.d.). Compared to other regions, the ASEAN is behind in adoption GIPS. The Americas continue to dominate the adoption of GIPS globally, with more than 81% of firms claiming compliance from the region. Europe, the Middle East and Africa (EMEA) account for 14%, and

the remainder are from the Asia Pacific (Grover, 2016). Grover (2016) further reported that for the 2015 notification period, only 78 of 1,621 firms claiming GIPS compliance were based in the Asia Pacific region (not down to the ASEAN as no breakdown figures are available). This number increased for 2016 with 94 out of 1,608<sup>1</sup> firms claiming compliance in the region (Grover, 2017). The CFA Institute indicates that these figures reveal an opportunity for stronger outreach efforts in the EMEA and Asia-Pacific (including ASEAN) where the GIPS standards are currently under-utilised.

There are three possible reasons as to why the ASEAN is under-represented in GIPS adoption, as discussed in the literature. The first relates to the investment firm scale in the region. Many of the local asset managers are classified as small scale. Investment firm scale is measured by assets under management (AUM), which is recognised as a limiting factor for GIPS adoption. The relationship between GIPS adoption and investment firm scale is evidenced in Healey's 2014 report, with the survey indicating that 40% of boutique investment managers agreed that the compliance with regulation becomes the biggest obstacle to expand their client base. The second reason may be due to a limited number of global firms locating their headquarters in the region (usually only having their "Asia branch" offices located in Singapore or Malaysia). The third reason is attributed to the fact that some developing countries of the ASEAN (that is, Cambodia, Laos, and Myanmar) have minimal exposure to international standards due to their immature legislation environment (Grover, 2016, 2017). Grover (2016) also argued that 2015's figures provided a starting point for conversations between GIPS and the local regulators, and the further promotion of standards in the region. With the scope for new adoption in the ASEAN countries, it would be valuable for the research to help understand what might influence this adoption process.

Moreover, it is notable that the ASEAN was relatively unscathed from the 2007–09 Global Financial Crisis due to the measurements taken after the Asian Financial Crisis of 1997–98 (Chia, 2011). This phenomenon subsequently encouraged economic policymakers and promoted investment benefits in the region. In March 2012, the ASEAN Investment Area (AIA) Council introduced the ASEAN Comprehensive Investment Agreement (ACIA), setting out the strategic objective as a "transparent and competitive investment environment" (ASEAN, n.d.a), which has resulted in many innovative projects with the unique feature of all being underpinned by the theme of inclusion in the investment management field. This social context would encourage and foster the adoption of global voluntary standards for investment management and make the ASEAN a fitting case/region for research on the adoption of GIPS.

#### 3. Theoretical framework for sustainability reporting

#### 3.1. Institutional theory and its representativeness of social pressure

Institutions are defined as "*regulative, normative and cognitive structures and activities that provide stability and meaning for social behaviour*" (Scott, 1995, p. 33). Institutional theory concentrates on the crucial question of why organisations in a field, over time, generally look/act the same (Miles, 2012). At the beginning of the organisational lifecycle, there is significant variety and form; however, there would be homogeneity in the organisational configuration and practices in the mature stages (DiMaggio & Powell, 1983). Early institutional scholars recognised the notion as institutional isomorphism, whereby processes and structures of organisations evolve and become similar to one another (DiMaggio & Powell, 1983; Meyer & Rowan, 1977). They suggested that organizations are homogeneous in composition because they need to gain a social mandate or legitimacy, and they do this by adhering to social expectations. Organisations not only begin to look alike but also respond similarly to regulatory, normative pressures and situations of uncertainty (DiMaggio & Powell, 1983).

According to DiMaggio and Powell (1983), the diffusion of ideas, standards and recommended structures facilitate the process of institutional isomorphism. Their framework covers the mechanisms through which institutional isomorphism change occurs (Boxenbaum & Jonsson, 2008). This approach involves testing the isomorphic pressure(s) to account for the diffusion of a standard or practice. The theory posits that three change mechanisms make institutional isomorphism happen by exerting pressure on organisations: coercive, mimetic and normative.

#### 3.2. Coercive isomorphism

Coercive isomorphic change is the result of regulative organisational elements that emphasize compliance as the basis for legitimacy. It applies when new policies, government mandates and financial reporting requirements are implemented. DiMaggio and Powell (1983) defined coercive isomorphism as resulting from formal and informal pressures exerted on organisations. Pressures from stakeholders and expectations from society persuade organisations to conform (Palthe, 2014; Peng, 2002). Despite empirical studies have been divided on the variables used to represent coercive isomorphic pressure (Lasmin, 2011), researchers agree in general that the adoption of global policies and standards is a kind of coercive isomorphic change (Henisz et al., 2005; Weber et al., 2009).

#### 3.3. Mimetic isomorphism

The mimetic isomorphic change focuses on cultural legitimacy. Organisations face uncertainty in times of making a significant change which could result from inadequately understood technologies, ambiguous goals, and unclear relationships (Henisz et al., 2005). In this case, shared models and beliefs are invoked and preferable (Palthe, 2014). An organisation may seek credibility by imitating the entities perceived as the most successful and prominent in their field (DiMaggio & Powell, 1983). By replicating those organisations that have already attained cognitive legitimacy, new entrants can also be deemed legitimate (Suchman, 1995).

Mimetic pressure was divided into three classifications: frequency-based, trait-based or outcome-based. Frequency-based imitation refers to the influence generated when a large number of organisations has adopted a structure or practice in the population. Trait-based imitation is when organizations look to replicate highly successful peers. Outcome-based imitation occurs following evaluation of the strategy of others. Outcome-based is limited to highly visible events, such as signals from the capital markets (Henisz & Delios, 2001).

#### 3.4. Normative isomorphism

Normative isomorphic change is associated with the immediate environment of the organization and moral legitimacy. Compliance with familiar structures and policies of a field stems from professionalism (DiMaggio & Powell, 1983). Examples of change mechanisms include networks that span the organisational area, inter-hiring between firms, andalso educational accreditation. Over time, professionals within a field, with a similar knowledge base and network, address issues in the same way (Palthe, 2014). Suchman (1995) defined the pursuit of professionalism as organisations "linking their activities to external definitions of authority and competence". The moral values imposed by a culture upon organisations also influence the decisions regarding what is an acceptable strategy (Martinez-Ferrero & Garcia-Sanchez, 2017, p. 589). Therefore, adherence to widely respected ethical standards and codes of best practice can have a similar effect and impact how an organisation operates (DiMaggio & Powell, 1983).

The claim to a supply of knowledge and technical specialists is a powerful source of credibility and a driver of normative isomorphic change identified by scholars (Hopwood, 2000; Power, 2003). Hopwood (2000) suggested that the adoption of standardised financial accounting practices or similar is not only functional but also symbolic of modernity and expertise. Furthermore, evidence supported a correlation between the enhancement of an organisations status and the membership of professional bodies and expert panels of the senior management (Abbott, 1981; Suchman, 1995).

In summary, pressures to make substantial changes on a specific process that results in a behavioural outcome in an organisation, a sector, a country, or a region, through political and legislative influence (coercive), replicating success (mimetic) and standardized professional norms and benchmarks (normative). All three isomorphic change mechanisms are therefore characteristic of a form of social pressure.

Until this point of the section, we have demonstrated social pressure can be represented by institutional isomorphism. In the next section, we move to develop hypotheses to test data based on empirical evidence on the relationships between social and economic pressures, respectively, and the adoption of GIPS.

#### 4. Empirical literature review and hypotheses development

#### 4.1. Social pressure and the adoption of GIPS

Researchers debate the concept of institutional context and refer to it as "rules, norms and ideologies" (Meyer & Rowan, 1977, p. 84), a regulatory framework of government and professional bodies (Meyer & Scott, 1983), or social setting and symbolic/cultural influence (Tolbert, 1985). However, it generally relates to broader social understandings and associated pressures, as well as the adaptive behaviour by organisations to secure an appropriate "fit" to the environment facing and the rationale behind it.

In the context of voluntary standard adoption, empirical studies are observed more on, for example, financial reporting standard but scarce on investment management standard such as GIPS. Arguably, they should share common ground in terms of globally accepted professional standards. These studies established a positive relationship between social pressure and the adoption of voluntary standards in financial reporting, e.g., Lasmin (2011) presented a correlation between all three isomorphic change mechanisms and the adoption of the IFRS. Similarly, a positive relationship between coercive, mimetic and normative pressures and assurance of sustainability reports is found in Martinez-Ferrero and Garcia-Sanchez (2017). They argued that voluntary adoption acts as a legitimisation tool; that is, countries that have a more significant legal system, and cultural development, are more likely to adopt voluntary standards. In contrast, research also acknowledged that despite coercive, mimetic and normative pressures drive compliance with IFRS, "a logic of resistance" was identified to the introduction of new standards (Maroun & Van Zijl, 2016) because new requirements are not perceived as competent. Therefore non-compliance by some stakeholders having negative sentiment is inevitable.

Although the GIPS Standards have been in circulation since 1999, less empirical evidence exists on GIPS studies. It could argue that the standards are now in the latter stage of the diffusion lifecycle since part or all of 23 of the 25 largest asset management firms globally claim compliance (Kent, 2016). In other words, this suggests that social pressure on the adoption of GIPS matured. For instance, the CFA Institute, as with any standard-setting group, is required to be aware of the social pressures influencing their body of work. At a country level, social pressures comprise coercive, mimetic and normative forces. Coercive force involves local organisations to develop and promote GIPS in partnership with the CFA Institute (Langevoort, 2014)<sup>2</sup>; mimetic force implies that the size of capital markets that represents the openness of the country's economy (Touron, 2005), which in turn may influence a country's willingness to comply with GIPS or replicate the processes of a perceived prosperous country; and normative forces are presented based upon the levels of higher education and the professional networks in the region. It can argue that the greater the influence of social pressures, the higher the adoption rate of GIPS. From this reasoning, the first hypothesis proposes:

**Hypothesis 1:** Social pressures, represented by coercive, mimetic and normative isomorphic change predictors, will exhibit a positive relationship with the adoption of GIPS.

#### 4.2. Economic pressures and the adoption of GIPS

Economic pressures define as changes in the business environment, such as the globalisation of markets and competition (Pirinen, 2005). A positive growth rate in GDP is a signal of expansion and good economic health. Increases in foreign direct investment, as well as domestic companies' financing activities abroad, also serve as an indicator of a countries credibility and financial position

(Lasmin, 2011). At the firm level, there is pressure to maximise the wealth of its shareholders, measured by the rate of the earnings per share, with the financial decisions being influenced by factors that would increase the net present worth of a company's profits (Rappaport, 2006).

For many standard-setting bodies, economic pressures often see to be the driving force of dissemination of the standard (Martell, 2010). Pirinen (2005) examined the adoption of International Accounting Standards (IAS) in Finland and found that economic pressures were the leading cause of the change in financial reporting in line with the international practice since the end of the 1970s, supplemented by social pressures. Another study carried out by Chi et al. (2013) in Taiwan revealed a compelling case: before 2001, under the Taiwan Companies Act, firms with contributed capital above a certain threshold had to produce financial reporting. This mandatory requirement was rescinded in 2001, leaving the voluntary reporting as an incentive to meet listing rules at the two local stock exchanges. The result shows that a continuation of financial reporting post-2001 correlated with the desire to obtain external financing—an excellent example of economic pressure impacting on the adoption of voluntary standards. Additionally, studies also found, while economic incentives are necessary, the quality and usefulness of reporting standards adoption would be high (Ball et al., 2003).

In the professional sector, Greenwood et al. (2002) studied accounting firms and concluded that by framing the profession, as being "under threat", it was portrayed that the way to stay competitive and meet new demands was to accept change by compliance with new standards. Taking investment management profession as an example, while it is not currently in the midst of a financial crisis as seen in 2007–09, the lack of transparency and risks control continue to call out as a barrier to trust (Edelman, 2016). This, in turn, has an economic repercussion for the sector, because asset owners would think twice before placing their investments. As such, it might be a good strategy for investment management firms to adopt best practices and seek compliance with standards such as GIPS to acquire new and retain existing business. This perspective can lead to the second hypothesis:

Hypothesis 2: Economic pressures demonstrate a positive relationship with the adoption of GIPS.

#### 4.3. Social vs. economic pressures and the adoption of GIPS

Our hypotheses 1 and 2 discuss the relationship between social, economic pressure respectively and the adoption of GIPS. This relationship is universal. We now would like to explore further to see social or economic pressure, which has more influence than the other in the adoption of GIPS. It could argue that the relationship should discuss in the context of a country or region in a certain period. Our case is ASEAN.

We have noticed that most of the previous studies on ASEAN regions have addressed issues on a country level, for example, Widyaningsih et al. (2019) study about the impact of audit rotation on audit quality in Indonesia. Adi et al. (2016) analysed the quality of performance report for local governments in the same country, while Nguyen's (2015) study addresses the lending rate in Vietnam following the financial crisis. Nor (2016) discusses the nature of tax planning in Malaysian multinational companies.

Despite both social pressure and economic pressure having a positive relationship with the voluntary adoption of global professional standards, research revealed in a certain period, and the two forces could conflict, e.g., Lasmin (2011) concluded that country adoption of IFRS is pressured more by social legitimacy in the form of isomorphic change than by economic pressures. The tension between economic and social influences can even provide insight into how global standards have been developed (Rodrigues & Craig, 2007).

In the context of ASEAN, the ASEAN Economic Community (AEC) was established in 2015, aiming to achieve an integrated and cohesive regional economy. The leaders of the member states

adopting the AEC Blueprint 2025 (ASEAN, n.d.b). The framework sets out steps to establishing a competitive single market and the free movement of goods, services, capital and investment between member countries. Studies agree that the ASEAN economy has experienced significant growth since 2000, embracing globalisation and attracting foreign investment (Vinayak et al., 2014; Yean & Das, 2015). However, critics noted that the AEC integration timetable has been a struggle for the region, with many objectives unfulfilled from the original 2007 plan and carried forward in the Blueprint 2025 (Lehmacher, 2016; Yean & Das, 2015). Some authors argued that the region's integration requires more open to the outside countries (Pangestu & Ing, 2016), such as establishing a partnership with non-ASEAN countries (Chen et al., 2017). One reason suggested for the slow progress is that regional responses to economic influence could only proceed with the consensus of all ten national governments (Lehmacher, 2016). A shared vision is difficult to reach because each ASEAN country is very different and have dissimilar priorities. For example, Indonesia is a member of the G20 representing nearly 40% of the region's economic output. In contrast, Myanmar is considered a frontier market that is emerging from decades of military control and civil war (Vinayak et al., 2014). Similarly, it could argue that external economic influences would have limited impact, due to the lack of consensus on the appropriate reaction from the region as a whole. Yean and Das (2015) suggested that resolution of conflicting domestic issues was a prerequisite to deeper integration and subsequently a cohesive regional response to economic pressures.

In contrast, the region's response to social pressures displays a strong cohesion, in the spirit of the ASEAN Way that has origins from tradition and culture because a key objective of the Socio-Cultural Community is to preserve a regional identity. The ASEAN Regional Forum plays a role in facilitating international relations and coordinating a single holistic response from the ten members (Laurence, 2007). Evidence shows that those countries with more power in the global system, or possessing high status have shaped the policies for countries that are less powerful or considered less legitimate (Gilpin, 1987). This explanation could apply to Singapore and Indonesia, in so far as other GIPS adopting countries influenced them in the investment management global community such as the UK and Japan. Likewise, within the ASEAN region, it could also be proposed, that the GIPS adopting countries will, in turn, exert the primary influence on the nonadopters. This inference leads to the argument that social pressures more impact the ASEAN region because the consensus among the member states is more prevalent than in response to economic pressures. From these discussions, the third hypothesis developed:

**Hypothesis 3:** Social pressures have more considerable influence than economic pressures in a country's adoption of GIPS within the ASEAN region.

#### 4.4. Social pressure components and the adoption of GIPS

In the studies of social pressure representing by coercive, mimetic and normative isomorphism, all three processes are embedded in different origins of how the behaviour of institutions diffuses. It suggests researchers do not give equal weighting to all three components, whereas being selective in interpretation, e.g., Tolbert and Zucker (1983) tested the influence of coercive pressure against mimetic pressure and found coercive pressure to be more effective in the spreading of new administration reform. In contrast, Lasmin (2011) confirmed that all three of the change mechanisms jointly influence accounting standards adoption internationally, but insufficient evidence supports an order of significance. The adoption of accounting standards by US firms highlighted normative pressure as the most important change mechanism (Mezias, 1990). In the study of Martinez-Ferrero and Garcia-Sanchez (2017), normative pressure was also the factor that provided the most significant explanatory power in the voluntary assurance of sustainability reports and followed by coercive pressure as a secondary stimulus. These results indicate that the evaluation of each social pressure component's influence on voluntary standard adoption, such as GIPS also needs to link to the institutional environment in the adoptive country or region.

Some studies identified that coercive pressures are likely to lead to ceremonial adoption of a structure or process because the adoption is not backed by locally embedded skill or motivation (Weber et al., 2009). Substantive adoption is likely to occur when internal knowledge and experience support the development and refinement of a practice (Meyer & Rowan, 1977) while professional networks can act as transmission channels for organizational models and technical knowledge (Owen-Smith & Powell, 2008). This finding is consistent with a primary objective of the GIPS Executive Committee, i.e. to partner with local sponsoring organisations, to promote expertise further and educate professionals on the standards. This driver of professionalism is characteristic of normative pressure. Concerning the context of ASEAN, we argue that normative pressure plays more roles than coercive and mimetic pressure. The evidence followed regarding the prospect of regional education, and the foster of professionalism and networks, and the increase of investment management research in the region can support our argument.

The initiation of ASEAN Vision 2020 calls for investments in the development of a knowledge economy. A case study of higher education in Malaysia and Thailand found that university interviewees placed significant importance on the international university rankings (Chapman & Chien, 2014). High rankings view as a way to attract top talent, foster professionalism and earn global respect. Approximately one-fifth of international students in Malaysia and Thailand are from ASEAN member states. Based on the assumption that graduates will stay in the region, it follows that the influence of educated professionals and networks will exert normative pressures on organisations to keep current with standards.

The CFA Institute reported that in the wider Asia-Pacific region, 11 of the 19 local member societies requested brand campaign funding (CFA Institute, 2016). In the ASEAN region, brand campaign initiatives planned for cultures in Indonesia, Philippines, Singapore and Malaysia. The campaigns based on out-of-home, digital and print advertisements, e.g., posters in the local commuter train stations and Facebook competitions. The Global Brand Campaign aims to deepen the understanding of the professionalism agenda and position the CFA Institute as the professional body for the investment management industry (CFA Institute, 2016). Research gathered from Scopus (a multi-disciplinary database of journals and conference papers) revealed a growth in publications in economics, econometrics and finance for the region (Halevi et al., 2014), with a growth rate of 14% in terms of publications for the period 1997 to 2012, the discipline ranked joint 5th out of 26 subjects.

This evidence shows a strong emphasis on professionalism, education and networks of talent in investment management-related subjects, as an essential influence that would support the adoption of GIPS. Besides this, there is little indication of a leading coercive pressure from regulators or capital suppliers concerning GIPS adoption. Also, there is no example of a standout investment manager in the region that is championing GIPS and would act as a motive for mimesis. Therefore, normative pressure appears to rank the highest, of the three isomorphic change mechanisms in the region with regards to GIPS adoption. Our final hypothesis arrives with this line of reasoning:

**Hypothesis 4:** Normative pressures are more influential than coercive and mimetic pressures in the adoption of GIPS in the ASEAN region.

#### 5. Research design

The primary objective of the paper is to examine how social and economic pressures impact the adoption of GIPs in the ASEAN region. The data used covers all 10 ASEAN countries spanning 17 years (1999–2015) for 170 country-year observations for each variable. The data collected from the World Bank database and the Chartered Financial Analyst (CFA) Institute. An effort was made to ensure the accuracy of the data before the analysis, various sensitivity analysis and control variables were added to secure the stability of results.

The 17 year period of the study allows the researcher the flexibility to run the study while minimising bias in comparison, due to the different timings of adoption, or expression of interest for different countries.

#### 5.1. Data description

The dependent variable (the adoption status of the GIPS) collected from the CFA institute online records. We measure it as a binary variable, where zero is "no adoption" for the country-year observation, and one indicates "a full adoption" or "expression of interest".

Table 1 demonstrates three levels of adoption (full adoption, non-adoption and expressed interest of adoption). We use only two levels in the analysis because the expression of interest is a transition period before full adoption for almost the case of the full adopters. The information in Table 1 was gathered on 1 January 2017 but data from previous years show a different pattern, as revealed in the frequency analysis of the adoption status.

Unlike Lasmin (2011), we do not want to limit our study to a single period as we want to reflect the behaviour over the entire period of the study. Moreover, to the best of our knowledge, this is the first study to be conducted in this region. Our aim is not limited to testing hypotheses, but to present to readers more information for the whole period in this region. Therefore, we generated a frequency analysis for the adoption status using three levels from 1999 to 2015 for the entire country-year observations.

The results reveal that the frequency (percentage) of full adoption status as 14 (8.2%), expressed interest as 38 (22.9%), and no adoption as 117 (68.8%). When rearranging the data per year, we can see that full adoption began in 2007, and for all four full adopter countries. The process was preceded by the expression of interest for different periods, beginning in 1999.

We argue that expressing interest led eventually to full adoption; this argument is valid during the period of the study for four out of five countries (except Malaysia). Meanwhile, in Brunei Darussalam, interest expression was made in 2014; thus leaving only a short period in which to convert to full adoption during the study.<sup>3</sup> Therefore, to facilitate better representation and in light of the results, we retain the adoption status as a binary variable on two levels. "Zero" is non-adopters having 117 observations representing 68.8% and "One" is adopter (fully or partially; expressing interest) having 53 observations representing 31.2%. This is the dependent variable (binary variable). The description of the other variables is to address in the following sub-section.

Table 2 presents the proxies and measurements of each variable and the sources of data. As indicated previously, all proxies have utilised in the literature. Further discussion about each proxy can see during the analysis of the results; a few more control variables were added to the model to capture any other impact from other resources and minimise the misspecification. Those variables will discuss in greater detail in the robustness analysis.

Table 1. ASEAN countries and	their GIPS adoption status as a	t 1 January 2017
Full Adopter	Expressed Interest	Non Adopter
Indonesia	Brunei Darussalam	Cambodia
Philippines	Malaysia	Lao PDR
Singapore		Myanmar
Thailand		Vietnam

Source: GIPS (n.d.)

Table 2. Summary	y of variables			
Adoption influence	Isomorphic Change Mechanism Proxy	Proxy/Variable	Measurement	Source
Adoption Status	N/A	Adoption Status of GIPS	Non Adopters 0 Adopters (Partial/ Full) 1	CFA Institute
Social Pressure	Coercive	Development Assistance	Net Official Development Assistance (ODA) received per capita	World Bank
Social Pressure	Mimetic	Stock Market	Market capitalization (MC) of listed domestic companies (% of GDP)	World Bank
Social Pressure	Normative	Professional Membership	Number of Chartered Financial Analyst (CFA) Charter-holders	CFA Institute
Economic Pressure	N/A	GDP Growth	Gross Domestic Product (GDP) growth (annual %)	World Bank

#### 5.2. Analytical model and supporting statistics

Alongside testing the hypotheses, we intend to provide readers with additional analyses to reveal more indicators to help future studies and enhance our contribution to the literature. The primary model runs as Logistic Regression testing for the hypotheses to address research objectives. Meanwhile, other analysis methods (e.g., description, correlation and Cross Tabs) adopted to provide supportive statistics to the main Binary Logistic model. After running the model, robustness tests are performed to ensure the trend of results will remain the same.

Given the fact that the response (Dependent) variable is a binary variable (0,1), Logistic Regression model is justified (Field, 2013, p. 761).<sup>4</sup> It is worth noting that as logistic require the dependent variable to be categorical which in our case is binary, it places no restrictions on the nature of the independent variables. Different measurement scales can be combined within the model among the independent variables (Field, 2013).

$$Logit_{\pi_i} = \beta_0 + \beta_1 ODA + \beta_2 MC + \beta_3 CFA + \beta_4 GDP$$

Where:

 $Logit_{\pi_i}$  = The logit of the probability of the adoption.<sup>5</sup>

 $\beta_0$  = The intercept of the regression equation.

ODA = Net Official Development Assistance received per capita.

MC = Market capitalisation as a percentage of Gross Domestic Product (GDP).

CFA = Number of CFA Charter-holders.

*GDP* = GDP growth Annual Percentage.

 $\beta_1, \beta_2, \beta_3, and \beta_4$  = The parameter estimates of the predictor variables; log-odds units.

(1)

#### 6. Empirical results and discussion

#### 6.1. Descriptive data

Table 3 presents mean, median, and standard deviation for the main variables. With regard to adoption status, the dependent variable in the model, the mean is 0.31, which confirms that there are 53 adoption observations out of the 170 valid observations, representing 31%. The median is 0 since more than half of the study population did not adopt GIPS during the study period; we anticipate this percentage will decline in the future based on the data trends. While this result needs to be treated with caution since the variable is binary and there is a time lag in the adoption of GIPS, the majority remain in the non-adopter category. However, we computed the percentage of the adoption including the year 2016 and found that the adoption status increased to 33%. Moving forward, we expect the rate to grow based on the trend of the figures of this variable. Thus, for future studies, it will be worth observing how our results would compare.

As a reflection of the variations among ASEAN countries, there are variations in most of the study variables. For example, Market Capitalisation varies among ASEAN countries, generating a high variance. Moreover, the number of CFA varies significantly among countries ending in very high variations. This suggests adopting a technical method of creating quantiles to reduce the variability while retaining the nature of the variable. This approach has been taken in several studies. For example, in her seminal paper, Dechow (1994) converted different variables into quantiles and ran the model per quantile. This issue has been addressed later when in the model analysis.

#### 6.2. Correlation and logistic regression

As a further step of exploring the data, we have generated a Pearson and Spearman correlation table with the adoption status. Table 4 presents the correlation coefficients for both Person and Spearman coefficients with the adoption status.

Table 3. Descripti	ve statistics for the	e main variables		
Variables	Valid Observations	Mean	Median	Std Dev.
Adoption Status	170	0.31	0.00	0.46
ODA	170	16.00	5.26	22.22
MC	170	50.41	7.17	71.87
CFA	170	289.69	31.50	669.85
CFA Quantiles	170	2.98	3.00	1.44
GDP Growth	170	5.83	6.01	3.35

Table 4. Correlation coefficien	t between adoption status and	independent variables
Variables	Adoption Status (Pearson)	Adoption Status (Spearman)
ODA	-0.44**	-0.49**
MC	0.73**	0.76**
CFA	0.56**	0.73**
CFA Quantiles	0.73**	0.73**
GDP Growth	-0.18**	-0.19**

\*\*Correlation is significant at the 0.01 level (2-tailed).

\*Correlation is significant at the 0.05 level (2-tailed).

From the results presented in Table 4, while both figures generated by Pearson and Spearman are having the same trend, there is a noticeable difference for the number of character CFA holders, reflecting the high variation, as detailed in Table 3. Closer figures are revealed for the two tools when we have created a quantile variable from the raw figures, as depicted in the row below the same variable in Table 4. We intend to use the quantile version of CFA holders as the normative proxy since it demonstrates more excellent stability than raw figures. The model will be run based on the original version for comparability in the robustness analysis section.

Correlation figures reveal a significant relationship with the adoption status; it should be noted that the sign was unexpected for two variables (ODA and GDP Growth). However, caution needs to be exercised here as correlation figures should not be used to suggest causality or as an indicator of the relationship direction since other factors may have led to such figures (Field, 2013, p. 270).

Logistic Regression is used to test the hypotheses and explore the main factors that may explain the adoption of GIPS. Three stages are undertaken: (1) testing multicollinearity problem; (2) testing the significance of the model, and (3) testing the coefficients of the independent variables in the model as a tool for testing the corresponding hypotheses.

Table 5 presents the model using Logistic Regression. It is divided into four panels (see explanation shortly). However, before testing the hypotheses based on the model outcomes, a multicollinearity problem must be examined. The figures required to detect such an issue are presented in Table 5.6

If tolerance value is less than 0.10, which means the Variance Inflation Factor (VIF) is more than 10, there will be a cause of concern for multicollinearity problem (Myers, 1990 as cited in Field, 2013, p. 795). In Table 5, all tolerance values are higher than 0.1 and even higher than 0.2. This means VIF amounts are less than 10, and even less than five. This shows there is no serious multicollinearity problem that may affect the model of the study outcomes. In other words, the interactions among independent variables are not strong enough to create noise that leads to model misspecification.

After examining the multicollinearity and ensuring there is no serious problem in the model, we can proceed to examine the overall model before exploring each hypothesis in the same order of the hypotheses development section. 95% confidence level will be applied (a typical in Social Sciences studies). However, any results remain significant at 90% for weaker outcomes, or 99% for more reliable results will be commented on.

Table 6 presents the final version of the Logistic Model, separated into various panels for simplicity of discussion.

indicator		y to detect any matteotimeanty
Variables in the Model	Coefficients <sup>a</sup> : Co	llinearity Statistics
	Tolerance	VIF
ODA	0.667	1.500
MC	0.348	2.870
CFA Quantiles	0.283	3.529
GDP Growth	0.782	1.279

Table 5. Tolerance statistic and variance inflation factor (VIF) to detect any multicollinearity

<sup>a</sup>Dependent Variable: Adoption Status for GIPS

Table 6 Locietic vouve	ccion autout					
Panel A: Omnibus Tests of	Model Coefficients					
Chi-square			df		Sia.	
133.210			4		00000	
Panel B: Model Summary						
-2 Log likelihood			Cox & Snell R Square		Nagelkerke R Squa	re
78.762 <sup>a</sup>			0.541		0.760	
Panel C: Classification Tab	ep	_		-		
Observed				Predict	ed	
			Adop	tion Status	Percentag	e Correct
			Non Adopter	Adopters: Partially or Fully		
Adoption Status	Non Ad	opter	106	11	06	.6
	Adopters: Part	ially or Fully	12	41	77	7.
Overall Percentage					86	.5
Panel D: Variables in the E	quation <sup>c</sup>					
<sup>c</sup> Variables entered	В	Wald	Sig.	Exp(B)	95% C.I. f	or Exp(B)
					Lower	Upper
ODA	-0.032	0.895	0.344	0.969	0.907	1.035
MC	0.013	3.180	0.075	1.014	0.999	1.029
CFA Quantiles	1.757	11.191	0.001	5.793	2.070	16.215
GDP Growth	-0.132	0.982	0.322	0.876	0.675	1.138
Constant	-6.968	15.000	0.000	0.001		
<sup>d</sup> Estimation terminated at i <sup>b</sup> The cut value is.500.	eration number 7 because	parameter estimates cha	nged by less than.001.	-		

In Table 6, Panel A reveals that the overall test model is significant at 95% and above; to test the model, we look at Omnibus Test of Model Coefficients where the output generates Chi-Square value that equal 132.10 with Sig. (P-Value) 0.000. This indicates that the model is statistically significant as P-value is 0.000, which is less than 5%. From the results presented so far in Tables 5 and 6, Panel A. It illustrates that there is no significant multicollinearity problem and the model as a whole is highly significant. In Table 6, Panel B reveals that the Nagelkerke R-square, an equivalence of R-Square in the Logistic Regression, is equal to 0.76 (76%). While Panel C in the same table shows that the overall percentage (hit ratio) of the model correctly classifying the voluntary adoption of the GIPS is 86.5%. More importantly, the significance of the model and the absence of significant multicollinearity enable authors to use the model to test the study hypotheses as follows. Panel D of Table 6 presents the results required for testing the study hypotheses.

Regarding the first hypothesis, there are three proxies for coercive, mimetic and normative. In Table 6, Panel D shows that the P values for Net Official Development Assistance (ODA), Market Capitalisation (MC), CFA are 0.344, 0.075, and 0.001, respectively. This indicates, at our confidence level, only CFA as a proxy for the normative dimension is significant. The significant levels of coefficients are for MC and the number of CFA holders, while the net ODA was insignificant. Regarding the direction of the relationship, the two significant coefficients are both positive since the EXP (B) for MC and CFA are 1.014 and 5.793 respectively. Further, we have generated the 95% confidence interval (CI) for Exp (B) for all variables, and it lower and upper bounds are 1 and above for CFA; thereby indicating a positive relationship significant at 95% CL. Meanwhile, the lower bound of MC is close to one (0.999), but it is significant at 90% CL. In figures, the 95% CI for MC = [0.999, 1.029] while for the CFA [2.070, 16.215] and for net ODA [0.907, 1.035]. The lower bound of net ODA CI was less than one and equal 0.907, and this coincides with the Exp (B) for net ODA = 0.969 showing negative direction, but this cannot rely on since it is insignificant as explained earlier and the upper bound is more than one equal 1.035.

To conclude, the first hypothesis is valid for two indicators which are MC and CFA at 90% confidence level, and only one indicator which is CFA at 95% and even at 99% given its P-value is 0.001 while there is no significant evidence for the ODA due to the high P-value. This leads to the conclusion that there is a moderate impact for the MC variable as the proxy for mimetic while the normative proxy has generated the highest confidence level. However, we leave the comparison for the third hypothesis.

For the second hypothesis, the results presented in Table 6 reveal that the parameter for the GDP growth as a proxy for the economic dimension is insignificant, P-value = 0.184, which indicates that this is an insignificant result at 90% confidence level. Further, the Exp (B) for GDP less than one along with the lower bound of the CI that equal 0.675. While the figures suggest a negative direction for the relationship, this cannot be relied on given the fact the outcome for this variable is insignificant. In conclusion, there is no evidence to suggest that GDP as a proxy for the economic dimension could have a positive impact on the attitude of adoption or expressing interest of GIPS. Thus, this argument provides no empirical support for our study.

The remaining two hypotheses are an extension of testing the first two hypotheses. A further comparison between social and economic pressures in Hypothesis 3 will be carried out, and among social pressure components in Hypothesis 4.

Hypothesis 3 suggests that social pressures are expected to have greater association with the adoption of GIPS than economic pressure measured by GDP growth. The fact that two of three social power proxies were positively significant at 90% as discussed in Hypothesis 1, and the GDP is insignificant as revealed in Hypothesis 2 make this issue clear and lead to the conclusion that overall social pressures are stronger than economic pressure in influencing adoption of GIPS at the country level in the ASEAN. As GIPS relates primarily to financial statements of the companies, macroeconomic factors are less important compared with Capital Markets and professional dimensions as a proxy for mimetic and

normative factors, respectively. To further support this outcome, the amount of Exp (B) for GDP was the lowest compared with other variables; Panel D of Table 6.

Moving forward in the comparison analysis as indicated in Hypothesis 4, we argue that the normative factor is higher than the other two social pressure factors in influencing the adoption of GIPS. The results presented in Table 6, Panel D reveal a definite advantage to the normative pressure measured by the number of the CFA holders, the EXP (B) for CFA equal to 5.793, which is higher than the EXP (B) of MC that equal 1.014. Further, the significance level of the normative factor proxy is lower than the significance level of MC. Thus, the confidence level of CFA is higher. In this hypothesis, the CFA is significant at 90, 95, 99% based on its P-value that = 0.001 while the MC is substantial at 90% only since its P-value 0.075. This yields sufficient evidence to accept this hypothesis, which complies with the literature. Moreover, we can conclude that the normative pressure is a higher factor of social pressure. This also reveals the value relevance for having more professionals in the ASEAN countries and encouraging other countries with low levels of professionals to work harder to attract and support more candidates to become professionally qualified.

#### 6.3. Robustness analysis

Other studies used the rule of law as a control variable as a proxy for coercive (Hillier et al., 2011; Martinez-Ferrero & Garcia-Sanchez, 2017). Both studies have used this control variable while analysing the firm-level data, which is not the case in our study as the firm-level data for ASEAN start to be available only in 2015. This prevents authors from conducting studies at the firm-level. However, the same model as in Equation (1) was running again using the new control variable. The results suggest the same trend, so our model proves stable when controlling for the law factor.<sup>7</sup> Simultaneously, we have added income level as a control variable for the degree of national governance (Ramanna & Stellen, 2009). We argue that the degree of national governance affects a country's decision to adopt GIPS. Following Lasmin (2011), we used the income level variable from the World Bank as a proxy for this variable.<sup>8</sup> Results followed the same trend and are available from authors, although they are not reported. Further combinations have been undertaken like running the model using the raw figures of CFA holders, none of earlier conclusions was affected; the normative proxy is still the highest. Foreign Direct Investment, net inflows (% of GDP) as an alternative proxy for economic pressure (Lasmin, 2011), among others. Again, the trend of results is not affected; thereby proving the robustness of the results.

#### 7. Summary and conclusion

Based on the earlier sections, our findings can be summarised over the following two points:

Based on the earlier sections, our findings can be summarised over the following two points:

(1) the adoption rate of GIPS in the ASEAN region is at a moderate level, with 31 per cent in 2015, and 33% in 2016. Compared to the 40% adoption rate of GIPS currently in the EU (GIPS, n.d.), the development is reasonable, considering the gap of social and economic development between these two regions.

(2) Social pressures (measured by development assistance level, stock market maturity, and professional network) jointly have a greater impact than economic pressure (measured by GDP growth) in the adoption of GIPS. The findings partially support the results in Lasmin (2011) and Martinez-Ferrero and Garcia-Sanchez (2017), and confirm that the social pressures that affect the adoption of IFRS and Sustainability Standards also apply to the GIPS case. Within the social pressure components, normative dimension (CFA membership as the proxy) has the largest impact over the mimetic (MC as the proxy) and coercive (ODA as the proxy) dimensions. Coercive isomorphic change has the least influence for the adoption of GIPS.

As the first study on this topic in the ASEAN region, with rigorous research design and robust analysis, our paper contributes to the literature in three ways.

First, we initialised the research using an institutional theory framework in order to examine the influences on the adoption of voluntary standards for investment performance measurement by providing a clear picture about GIPS adoption in the ASEAN region—an overlooked area, from which, a distance can be identified with other regions (for example, the EU) that contain developed countries.

Second, our findings have identified both the most and least important components within social pressures with normative dimension as being more important than mimetic, while coercive is the least influential factor in the adoption of GIPS.

Third, we made a methodologic contribution by combining data from two different sources and presenting relevant results for current and future studies.

Our findings have important policy and managerial implications. On the one hand, policymakers in the ASEAN should focus on the reinforcement of legislation and regulation infrastructure in order to improve the political, economic, and cultural integration in the region as a whole, to support the implementation of international professional standards such as GIPS, and on the other hand, they should also recognise that the profession or professional networks can and have played a significant role in the promotion of these international standards, evidenced by our findings which indicate that normative isomorphic change has made a useful contribution to the adoption of GIPS.

The current study has limitations at regional level as a result of data availability. However, while an objective of the ASEAN is to become more integrated, it might be a good idea to replicate our study in a few years when adoption would be more widespread for non-adopting countries, and more established for the adopting countries. Moreover, it expects the professional (normative) dimension would remain dominant in the field, thus it is interesting to see the movement towards capital market dimension. Our study found that approximately 95 per cent of country-year observations in the adoption category have a capital market in their countries, and therefore we expect this variable to develop more influence in the future.<sup>9</sup> A firm-level study is needed once the data become available.

Furthermore, market-based research is recommended as we expect a more significant role in this dimension. Once this is proven, it would be excellent to see a capital market study in this field, once more data are available. Finally, we deem that it would be an excellent idea moving forward to integrate between country level and firm-level, once both studies are conducted separately. We regard our study as the base of the recommendations mentioned earlier, and this can be thought of as an additional contribution alongside our results.

#### Funding

The authors received no direct funding for this research.

#### Author details

Deborah Zelikson<sup>1</sup> Moade Fawzi Shubita<sup>2</sup> E-mail: m.f.shubita@leedsbeckett.ac.uk ORCID ID: http://orcid.org/0000-0002-1465-9047 Juniie Wu<sup>2</sup> E-mail: J.Wu@leedsbeckett.ac.uk

ORCID ID: http://orcid.org/0000-0002-6363-2705

- <sup>I</sup> Technology, Bny Mellon Investment Company, Leeds, UK. <sup>2</sup> Leeds Business School, Leeds Beckett University, Leeds,
- UK.

#### **Cover Image**

Source: Author

#### Citation information

Cite this article as: Adoption of global investment performance standards: Case of ASEAN, Deborah Zelikson,

Moade Fawzi Shubita & Junjie Wu, Cogent Business & Management (2020), 7: 1746169.

#### Notes

- 1. The total number decreased because approximately 150 firms that registered in 2015 missed the 2016 window due to a new annual renewal process coming into effect. Also, merger and acquisition activity led to the consolidation of some firms (Grover, 2017).
- 2. Examples include the CFA Society Local Charters as in the Philippines or the Investment Management Associations as in Singapore (GIPS, n.d.).
- 3. The variation within the same aroup is normal behaviour. For example, Shubita (2015) explores the impact of income smoothing on earning quality in the Gulf Corporation Council (GCC), one country in that region shows different behaviour than all others without driving the overall results.
- 4. This approach widely used in other studies when the dependent variable is categorical such as (Cho et al., 2014; Cowan et al., 2013; Herda et al., 2014;

Lasmin, 2011; Martinez-Ferrero & Garcia-Sanchez, 2017).

- Logistic Regression model the probability (P) of an even that is the Adoption of GIPS in our study. Thus, the extended form of the dependent variable is ln(<sup>p</sup>/<sub>1-n</sub>).
- 6. This problem occurs when there is a strong correlation between one or more of the independent variables. We will use two indicators as explained in the text to detect such a potential problem: Variance Inflation Factor (VIF) and tolerance statistics.
- The same trend does not indicate same figures; significant level remains within the same ranges of the first outcomes, so conclusions about the hypotheses stay the same. Detailed computer outputs are available from authors upon request.
- Income level in the World Bank database is four levels, ASEAN countries are distributed based on the countryyear observation as 58 (32.2%) low income, 61 (33.9%) lower-middle-income, 25 (13.9%) upper middle income, and 36 (20%) High-income economics. In the model, we created dummy variable two levels; zero for low level and one otherwise.
- 9. This figure arises from running cross tabs for adoption status against a dummy variable for the existence of the capital market. We are presenting this portion, as we believe it may be relevant for future studies, and we are happy to share any results, along with the data, with any researcher in the future.

#### References

- Abbott, A. (1981). Status and status strain in the professions. American Journal of Sociology, 86(4), 819–835. https://doi.org/10.1086/227318
- Adi, S., Martani, D., Pamungkas, B., & Simanjuntak, R. A. (2016). Analysis of the quality of performance report of the local government on websites: Indonesian case. *Cogent Business & Management*, 3(1), 1229393. https://doi.org/10.1080/23311975.2016.1229393
- ASEAN. (n.d.a). About ASEAN [Online]. Retrieved April 6, 2016, from http://www.asean.org
- ASEAN. (n.d.b) ASEAN Investment Area (AIA) council [Online]. Retrieved March 1, 2017, from http://www. asean.org
- Ball, R., Robin, A., & Wu, J. S. (2003). Incentives versus standards: Properties of accounting income in four East Asian countries. *Journal of Accounting and Economics*, 36(1), 235–270. https://doi.org/10.1016/j. jacceco.2003.10.003
- Boxenbaum, E., & Jonsson, S. (2008). Isomorphism, diffusion and decoupling. In G. Royston, C. Oliver,
  K. Sahlin, & S. Roy (Eds.), *The SAGE handbook of organizational institutionalism* (pp. 78–98). SAGE Publications.
- CFA Institute. (2014). Global investment performance standards 2010.
- CFA Institute. (2016, September). Global brand campaign continues to expand. CFA Institute Magazine, 20.
- Chapman, D., & Chien, C.-L. (Eds.). (2014). Expanding out and up: What are the system-level dynamics? Case study of Malaysia and Thailand. In UNESCO institute of statistics. Higher education in Asia: Expanding out, expanding up (pp. 37–48). UIS Publications.
- Chen, L., De Lombaerde, P., & Cuyvers, L. (2017). Markets matter: The potentialof intra-regional trade in ASEAN and its implications for Asian regionalism. Asian Economic Papers, 16(2), 1–19. https://doi.org/10. 1162/ASEP\_a\_00510
- Chi, W., Dhaliwal, D., Li, O., & Lin, T.-H. (2013). Voluntary reporting incentives and reporting quality: Evidence from a reporting regime change for private firms in

Taiwan. Contemporary Accounting Research, 30(4), 1462–1489. https://doi.org/10.1111/1911-3846.12003

- Chia, S. (2011). Association of Southeast Asian nations economic integration: Developments and challenges. *Asian Economic Policy Review*, 6(1), 43–63. https://doi. org/10.1111/j.1748-3131.2011.01178.x
- Cho, K. R., Huang, C.-H., & Padmanabhan, P. (2014). Foreign ownership mode, executive compensation structure, and corporate governance: Has the literature missed an important link? Evidence from Taiwanese firms. *International Business Review*, 23 (2), 371–380. https://doi.org/10.1016/j.ibusrev.2013. 06.005
- Cowan, A., Huang, C.-H., Padmanabhan, P., & Wang, C.-H. (2013). The determinants of foreign giving: An exploratory empirical investigation of US manufacturing firms. *International Business Review*, 22(2), 407–420. https://doi.org/10.1016/j.ibusrev.2012.05. 005
- Darus, F., Hamzah, E., & Yusoff, H. (2013). CSR web reporting: The influence of ownership structure and mimetic isomorphism. In: International conference on economics and business research 2013, Procedia – Economics and finance (Vol. 7, pp. 236–242). Elsevier Ltd.
- Dechow, P. (1994). Accounting earnings and cash flows as measures of firm performance: The role of accounting accruals. *The Journal of Accounting and Economics*, 18(1), 3–42. https://doi.org/10.1016/ 0165-4101(94)90016-7
- DiMaggio, P., & Powell, W. (1983). The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields. *American Sociological Review*, 48(2), 147–160. https://doi.org/10.2307/2095101
- Edelman, R. (2016). Edelman trust barometer Annual global study [Online]. Edelman. Retrieved March 25, 2016, from http://www.edelman.com/
- Evans, C. L. (2013, December 3). Why investment performance standards matter. *Market Integrity Insights* [Online blog]. Retrieved April 7, 2017, from https:// www.blogs.cfainstitute.org
- Field, A. (2013). Discovering statistics using IBM SPSS statistics: And sex and drugs and rock 'n' roll. SAGE.
- Friedland, R., & Alford, R. (1991). Bringing society back in: Symbols, practices, and institutional contradictions. In W. Powell & P. DiMaggio (Eds.), The new institutionalism in organizational analysis (pp. 232–263). Chicago Press Ltd.
- Gilpin, R. (1987). The political economy of international relations. Princeton University Press.
- GIPS. (n.d.) Country sponsors [Online]. CFA Institute. Retrieved March 23, 2017, from https://www.gips standards.org
- Greenwood, R., Suddaby, R., & Hinings, C. R. (2002). Theorizing change: The role of professional associations in the transformation of institutionalized fields. The Academy of Management Journal, 45(1), 58–80. https://doi.org/10.2307/3069285
- Grover, A. (2016, March, 23). 74% of top 100 global asset management firms claim GIPS compliance. *Market Integrity Insights* [Online blog]. Retrieved March 23, 2016, from https://blogs.cfainstitute.org/marketin tegrity/2016/03/23/74-of-top-100-global-assetmanagement-firms-claim-gips-compliance/
- Grover, A. (2017, February 6). Out of top 100 asset management firms globally, 85 claim GIPS compliance. *Market Integrity Insights* [Online blog]. Retrieved April 21, 2017, from https://blogs.cfainstitute.org/market integrity/2017/02/06/out-of-top-100-assetmanagement-firms-globally-85-claim-gipscompliance/

- Halevi, G., & Moed, H. (2014). International scientific collaboration. In UNESCO Institute of Statistics (Eds), *Higher education in Asia: Expanding out, expanding* up (pp. 79–92). UIS Publications.
- Healey, R. (2014, July 29). the new boutique asset manager: Transparent, efficient and accountable. Research report. Tabb Group LLC.
- Henisz, W., & Delios, A. (2001). Uncertainty, imitation and plant location: Japanese multinational corporation 1990–1996. Administrative Science Quarterly, 46(3), 443–475. https://doi.org/10.2307/3094871
- Henisz, W., Zelner, B., & Guillen, M. (2005). Marketoriented infrastructure reforms 1977–1999. American Sociological Review, 70(6), 871–897. https://doi.org/ 10.1177/000312240507000601
- Herda, D., Taylor, M., & Winterbotham, G. (2014). The effect of country-level investor protection on the voluntary assurance of sustainability reports. *Journal* of International Financial Management & Accounting, 25(2), 209–236. https://doi.org/10.1111/jifm.12018
- Hopwood, A. (2000). Understanding financial accounting practice. Accounting, Organizations and Society, 25 (8), 763–766. https://doi.org/10.1016/S0361-3682(00) 00021-0
- Kent, C. (2016, October 12). The evolution of the GIPS standards: Introducing GIPS 20: 20. Market Integrity Insights [Online blog]. Retrieved February 26, 2017, from https://www.blogs.cfainstitute.org
- Langevoort, D. (2014, November 1). Ethical mindfulness: A guide for new financial services professionals [Online]. CFA Institute. Retrieved April 11, 2017, from https://www.cfainstitute.org
- Lasmin. (2011). Accounting standards internationalization revisit: Managing responsible diffusion. *Procedia* - *Social and Behavioural Sciences*, *25*, 363–374. https://doi.org/10.1016/j.sbspro.2011.10.555
- Laurence, H. (2007). The ASEAN way and community integration: Two different models of regionalism. *European Law Journal*, 13(6), 857–879. https://doi. org/10.1111/eulj.2007.13.issue-6
- Lehmacher, W. (2016, May 31). The ASEAN economic community: What you need to know. *World Economic Forum* [Online]. Retrieved April 11, 2017, from https:// www.weforum.org
- Maroun, W., & Van Zijl, W. (2016). Isomorphism and resistance in Implementing IFRS 10 and IFRS 12. *British Accounting Review*, 48(2), 220–239. https://doi. org/10.1016/j.bar.2015.07.003
- Martell, L. (2010). The sociology of globalization. Polity Press.
- Martinez-Ferrero, J., & Garcia-Sanchez, I.-M. (2017). Coercive, normative and mimetic isomorphism as determinants of the voluntary assurance of sustainability reports. *International Business Review*, 26(1), 102–118. https:// doi.org/10.1016/j.ibusrev.2016.05.009
- Meyer, J., & Rowan, B. (1977). Institutionalized organizations: Formal structure as myth and ceremony. *American Journal of Sociology*, 83(2), 340–363. https://doi.org/10.1086/226550
- Meyer, J., & Scott, W. R. (1983). Organizational environments: Ritual and rationality. Sage Publications.
- Mezias, S. (1990). An institutional model of organizational practice: Financial reporting at the fortune 200. Administrative Science Quarterly, 35(3), 431–457. https://doi.org/10.2307/2393312
- Miles, J. (2012). Management and organisational theory [Online]. John Wiley & Sons Inc. Retrieved April 1, 2017, from http://lib.myilibrary.com.ezproxy.leeds beckett.ac.uk
- Myers, R. (1990). Classical and modern regression with applications (2nd ed.). Duxbury.

- Nguyen, C. V. (2015). The Vietnamese lending rate, policy-related rate, and monetary policy post-1997 Asian financial crisis. *Cogent Economics & Finance, 3* (1), 1007808. https://doi.org/10.1080/23322039. 2015.1007808
- Nor, S. A. W. (2016). Malaysian multinational companies (MNC): Permanent and temporary nature of tax planning. Cogent Business & Management, 3(1), 1248644. https://doi.org/10.1080/23311975.2016. 1248644
- Nyahas, S. I., Munene, J. C., Orobia, L., & Kaawaase, T. K. (2017). Isomorphic influences and voluntary disclosure: The mediating role of organizational culture. *Cogent Business & Management*, 4(1), 1351144. https://doi.org/10.1080/23311975.2017.1351144
- Orcalli, G. (2017). Market building through regional integration agreements: The EU and the ASEAN way. *Journal of Economic Integration*, 32(1), 160–192. https://doi.org/10.11130/jei.2017.32.1.160
- Owen-Smith, J., & Powell, W. (2008). Networks and institutions. In G. Royston, C. Oliver, K. Sahlin, &
  R. Suddaby (Eds.), The SAGE handbook of organizational institutionalism (pp. 596–623). SAGE Publications.
- Palthe, J. (2014, May 27). Regulative, normative, and cognitive elements of organizations: Implications for managing change [Online]. Sciedupress. Retrieved April 14, 2016, from https://mos.sciedupress.com
- Pangestu, M., & Ing, Y. L. (2016). ASEAN: Regional integration and reforms. Asian Economic Papers, 15(2), 44–60. https://doi.org/10.1162/ASEP a 00424
- Peng, D. (2002). Invisible linkages: A regional perspective of East Asian political economy. *International Studies Quarterly*, 46(3), 423–447. https://doi.org/10.1111/ isqu.2002.46.issue-3
- Pindado, Hillier, D., & De Queiroz, J., V. & De La Torre, C. (2011) The Impact of Country-level Corporate Governance on Research and Development. Journal of International Business Studies, 42(1), 76–98. https://doi.org/10.1057/jibs.2010.46
- Pirinen, P. (2005). Economic and normative pressures as drivers for the adoption of international accounting standards in Finland since 1976. European Accounting Review, 14(1), 213–235. https://doi.org/ 10.1080/0963818042000338979
- Power, M. (2003). Auditing and the production of legitimacy. Accounting, Organizations and Society, 28(4), 379–394. https://doi.org/10.1016/S0361-3682(01)00047-2
- Ramanna, K., & Stellen, E. (2009). Why do countries adopt International Financial Reporting Standards? In Harvard Business School Accounting & Management Unit Working, 09–102. https://papers.ssrn.com/sol3/ papers.cfm?abstract\_id=1460763#references-widget
- Rappaport, A. (2006, September). Ten ways to create shareholder value. Harvard Business Review [Online]. Retrieved April 2, 2017, 84(9), 66–77. https://hbr.org/ 2006/09/ten-ways-to-create-shareholder-value
- Rodrigues, L., & Craig, R. (2007). Assessing international accounting harmonization using hegelian dialectic, isomorphism and foucault. *Critical Perspectives on Accounting*, 18(6), 739–757. https://doi.org/10.1016/j. cpa.2006.02.007
- Scott, R. (1995). Institutions and organizations. Sage Publications.
- Shubita, M. (2015). The impact of income smoothing on earnings quality in emerging markets: Evidence from GCC markets. Journal of Accounting in Emerging Economies, 5(3), 299–324. https://doi.org/10.1108/ JAEE-04-2011-0011
- Suchman, M. (1995). Managing legitimacy: Strategic and institutional approaches. Academy of Management

*Review*, 20(3), 571–610. https://doi.org/10.5465/amr. 1995.9508080331

- Thampanya, N., Wu, J., Nasir, M., & Jia, L. (2020). Fundamental and behavioural determinants of stock return volatility in ASEAN-5 countries. *Journal of International Financial Markets, Institutions & Money.* Forthcoming.
- Tolbert, P. (1985). Institutional environments and resource dependence: Sources of administrative structure in institutions of higher education. *Administrative Science Quarterly*, 30(1), 1–13. https://doi.org/10.2307/2392808
- Tolbert, P., & Zucker, L. (1983). Institutional sources of change in the formal structure of organizations: The diffusion of civil service reform, 1880–1935. Administrative Science Quarterly, 28(1), 22–39. https://doi.org/10.2307/2392383
- Touron, P. (2005). The adoption of US GAAP by French firms before the creation of the international accounting standards committee: An institutional explanation. *Critical Perspectives on Accounting*, 16(6), 851–873. https://doi.org/10.1016/j.cpa.2003.08.011

- Vinayak, H. V., Thompson, F., & Tonby, O. (2014, May). Understanding ASEAN: Seven things you need to know [Online]. McKinsey & Company. Retrieved April 23, 2017, from http://www.mckinsey.com
- Weber, K., Davis, G., & Lounsbury, M. (2009). Policy, myth and ceremony: The global spread of stock exchanges, 1980–2005. Academy of Management Journal, 52(6), 1319–1347. https://doi.org/10.5465/amj.2009.47085184
- Widyaningsih, I. A., Harymawan, I., Mardijuwono, A. W., Ayuningtyas, E. S., & Larasati, D. A. (2019). Audit firm rotation and audit quality: Comparison before vs after the elimination of audit firm rotation regulations in Indonesia. *Cogent Business & Management*, 6(1), 1695403. https:// doi.org/10.1080/23311975.2019.1695403
- World Bank. (2016). World development indicators [Online]. Retrieved March 25, 2016. http://www. worldbank.org/
- Yean, T. S., & Das, S. (2015). The ASEAN economic community and conflicting domestic interests. Journal of Southeast Asian Economics, 32(2), 189–201. https:// doi.org/10.1355/ae32-2a



#### © 2020 The Author(s). This open access article is distributed under a Creative Commons Attribution (CC-BY) 4.0 license.

You are free to:

Share — copy and redistribute the material in any medium or format. Adapt — remix, transform, and build upon the material for any purpose, even commercially. The licensor cannot revoke these freedoms as long as you follow the license terms. Under the following terms: Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use. No additional restrictions

You may not apply legal terms or technological measures that legally restrict others from doing anything the license permits.

#### *Cogent Business & Management* (ISSN: 2331-1975) is published by Cogent OA, part of Taylor & Francis Group. Publishing with Cogent OA ensures:

- Immediate, universal access to your article on publication
- High visibility and discoverability via the Cogent OA website as well as Taylor & Francis Online
- Download and citation statistics for your article
- Rapid online publication
- Input from, and dialog with, expert editors and editorial boards
- Retention of full copyright of your article
- Guaranteed legacy preservation of your article
- Discounts and waivers for authors in developing regions

Submit your manuscript to a Cogent OA journal at www.CogentOA.com